Author's response to reviews

Title: Laparoscopic versus Open Left Lateral Segmentectomy

Authors:

Kirstin A Carswell (kirstincarswell@hotmail.com)
Filippos Sagias (filippos@doctors.org.uk)
Beth Murgatroyd (beth@londonwls.co.uk)
Mohamed Rela (mohamed.rela@kch.nhs.uk)
Nigel Heaton (nigel.heaton@kch.nhs.uk)
Ameet G Patel (ameet.patel@kch.nhs.uk)

Version: 2 Date: 8 February 2009

Author's response to reviews:

The original version has been amended due to a duplication in the uploading of Figure 1.